\begin{thebibliography}{10}

\bibitem{Ariola89}
Zena~M. Ariola and Arvind.
\newblock {P-TAC: A Parallel Intermediate Language}.
\newblock In {\em Proc. Functional Programming Languages and Computer
  Architecture}, September 1989.

\bibitem{Ariola91}
Zena~M. Ariola and Arvind.
\newblock {A Syntactic Approach to Program Transformations}.
\newblock In {\em Proc. Partial Evaluation and Semantics-Based Program
  Manipulation}, pages 116--129. ACM Press, September 1991.

\bibitem{Ariola94}
Zena~M. Ariola and Arvind.
\newblock {Properties of a First-order Functional Language with Sharing}.
\newblock CSG Memo 347-1, Laboratory for Computer Science, MIT, Cambridge, MA
  02139, June 1994.
\newblock To appear in Theoretical Computer Science, September 1995.

\bibitem{Arvind86}
Arvind and David~E. Culler.
\newblock {Dataflow Architectures}.
\newblock In {\em Annual Reviews in Computer Science}, volume~1, pages
  225--253. Annual Reviews Inc., Palo Alto, CA, 1986.

\bibitem{Arvind89}
Arvind, Rishiyur~S. Nikhil, and Keshav~K. Pingali.
\newblock {I-Structures: Data Structures for Parallel Computing}.
\newblock {\em {ACM Transactions on Programming Languages and Systems}},
  11(4):{598--632}, 1989.

\bibitem{Barth92}
Paul~S. Barth.
\newblock {\em {Atomic Data Structures for Parallel Computing}}.
\newblock PhD thesis, Laboratory for Computer Science, MIT, Cambridge, MA
  02139, March 1992.
\newblock Available as Technical Report MIT/LCS/TR-532.

\bibitem{Barth91}
Paul~S. Barth, Rishiyur~S. Nikhil, and Arvind.
\newblock {M-Structures: Extending a Parallel, Non-Strict, Functional Language
  with State}.
\newblock In {\em Proc. Functional Programming Languages and Computer
  Architecture}, pages 538--568. Springer-Verlag, 1991.
\newblock LNCS 523.

\bibitem{Hoare85}
C.A.R. Hoare.
\newblock {\em {Communicating Sequential Processes}}.
\newblock Prentice Hall International Series in Computer Science. Prentice
  Hall, Englewood Cliffs, NJ, 1985.

\bibitem{Haskell90}
P.~Hudak and P.~Wadler~(editors).
\newblock Report on the programming language {H}askell, a non-strict purely
  functional language ({V}ersion 1.0).
\newblock Technical Report YALEU/DCS/RR777, Department of Computer Science,
  Yale University, April 1990.

\bibitem{Hudak93}
Paul Hudak.
\newblock {Mutable Abstract Datatypes -or- How to Have Your State and Munge It
  Too}.
\newblock Research Report YALEU/DCS/RR-914, Department of Computer Science,
  Yale University, New Haven, CT 06520, December 1992.
\newblock Revised May 1993.

\bibitem{Johnsson85}
Thomas Johnsson.
\newblock {Lambda Lifting: Transforming Programs to Recursive Equations}.
\newblock In {\em Proc. Functional Programming Languages and Computer
  Architecture}, pages 190--203. Springer-Verlag, 1985.
\newblock LNCS 201.

\bibitem{Launchbury94}
John Launchbury and Simon~L. Peyton~Jones.
\newblock {Lazy Functional State Threads}.
\newblock In {\em Proc. Programming Language Design and Implementation}, pages
  24--35. ACM Press, June 1994.

\bibitem{Id90.1}
Rishiyur~S. Nikhil.
\newblock {Id Language Reference Manual Version 90.1}.
\newblock Technical Report CSG Memo 284-2, Laboratory for Computer Science,
  MIT, Cambridge, MA 02139, July 15 1991.

\bibitem{pH-doc}
Rishiyur~S. Nikhil, Arvind, and James Hicks.
\newblock {pH Language Proposal (Preliminary)}.
\newblock Circulated on the pH mailing list, September 1993.

\bibitem{PeytonJones93}
Simon~L. Peyton~Jones and Philip Wadler.
\newblock {Imperative Functional Programming}.
\newblock In {\em Proc. Principles of Programming Languages}, pages 71--84. ACM
  Press, January 1993.

\bibitem{Stoy94}
J.E. Stoy.
\newblock {The Semantics of Id}.
\newblock In A.W. Roscoe, editor, {\em A Classical Mind: Essays in Honour of
  C.A.R. Hoare}, pages 379--404. Prentice Hall, New York, 1994.

\bibitem{Berkeley-io}
Thorsten von Eicken, David Culler, and Klaus~Erik Schauser.
\newblock {Berkeley Id90 I/O proposal}.
\newblock Draft Memo, August 1991.

\end{thebibliography}

% -*- Mode: LaTeX -*-

\section{Introduction}
\label{sec:intro}

Data-parallel programming models, as embodied in *Lisp and CM-Fortran,
identify well-defined computation and communication phases that are
separated by barriers to ensure proper synchronization among the
various parallel activities. Some parallel machines like the CM5 and
Cray T3D even provide such barrier support in hardware.  In this
paper we study the use and implementation of barriers under a more
general, multi-threaded, MIMD programming model, in which
barriers need to be localized to specified subsets of
parallel tasks in order to provide fine-grain control over
synchronization and resource usage.

Our study is in the context of Id \cite{Id90.1} and pH \cite{pH-doc},
which are non-strict, implicitly-parallel, functional languages
extended with I-Structures \cite{Arvind89} and M-Structures
\cite{Barth91}. Barriers were introduced in Id to provide partial
sequencing of M-Structure operations.  However, their semantics were
not given precisely at that stage \cite{Barth92}.  Subsequently, pH
was designed to follow the precise definition of Haskell
\cite{Haskell90} combined with the eager evaluation model and the
non-functional extensions of Id. This prompted a formalization of
barrier semantics for both Id and pH. The first such attempt was made
in \cite{Stoy94}, both in terms of CSP \cite{Hoare85} and (for the
functional layer) denotationally.  This paper presents a different
formalization in terms of graph rewriting systems previously used to
describe the semantics of Id \cite{Ariola94}.

The rest of the paper is organized as follows. In \sec{eager}, we
discuss some important characteristics of non-strict, eager
computations. We describe the fully parallel evaluation model of Id
and pH, differentiating between the \ibx{result} and the
\ibx{termination} of an eager computation, and introduce the concept
of control regions.  In \sec{uses}, we introduce barriers informally
and present several examples of their use in controlling parallelism,
managing program resources, and providing fine-grain control over
scheduling of side-effects.

In \sec{kernel}, we describe the formal framework for barrier
semantics by introducing a simple kernel language with parallel
reduction semantics and a weak head-normal form operator (\lbx{W}).
In \sec{dataflow}, we describe the concept of termination and propose
the first semantics for barriers based on a purely eager, dataflow
graph execution model. This is sufficient for both resource
management and side-effect control, although it turns out to be a
little fragile with respect to program transformations.  \sec{frag}
highlights these shortcomings and identifies some ways to fix them.
In \sec{alter}, we present the second semantics for barriers,
attempting to overcome some of these shortcomings by mixing eager
evaluation with lazy identifier referencing.  Finally, \sec{conclude}
presents the summary and directions for future work.
